\documentclass[twocolumn,twoside]{article}
\makeatletter\if@twocolumn\PassOptionsToPackage{switch}{lineno}\else\fi\makeatother

\usepackage{amsfonts,amssymb,amsbsy,latexsym,amsmath,tabulary,graphicx,times,xcolor}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% Following additional macros are required to function some 
% functions which are not available in the class used.
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\usepackage{url,multirow,morefloats,floatflt,cancel,tfrupee}
\makeatletter


\AtBeginDocument{\@ifpackageloaded{textcomp}{}{\usepackage{textcomp}}}
\makeatother
\usepackage{colortbl}
\usepackage{xcolor}
\usepackage{pifont}
\usepackage[nointegrals]{wasysym}
\urlstyle{rm}
\makeatletter

%%%For Table column width calculation.
\def\mcWidth#1{\csname TY@F#1\endcsname+\tabcolsep}

%%Hacking center and right align for table
\def\cAlignHack{\rightskip\@flushglue\leftskip\@flushglue\parindent\z@\parfillskip\z@skip}
\def\rAlignHack{\rightskip\z@skip\leftskip\@flushglue \parindent\z@\parfillskip\z@skip}

%Etal definition in references
\@ifundefined{etal}{\def\etal{\textit{et~al}}}{}


%\if@twocolumn\usepackage{dblfloatfix}\fi
\usepackage{ifxetex}
\ifxetex\else\if@twocolumn\@ifpackageloaded{stfloats}{}{\usepackage{dblfloatfix}}\fi\fi

\AtBeginDocument{
\expandafter\ifx\csname eqalign\endcsname\relax
\def\eqalign#1{\null\vcenter{\def\\{\cr}\openup\jot\m@th
  \ialign{\strut$\displaystyle{##}$\hfil&$\displaystyle{{}##}$\hfil
      \crcr#1\crcr}}\,}
\fi
}

%For fixing hardfail when unicode letters appear inside table with endfloat
\AtBeginDocument{%
  \@ifpackageloaded{endfloat}%
   {\renewcommand\efloat@iwrite[1]{\immediate\expandafter\protected@write\csname efloat@post#1\endcsname{}}}{\newif\ifefloat@tables}%
}%

\def\BreakURLText#1{\@tfor\brk@tempa:=#1\do{\brk@tempa\hskip0pt}}
\let\lt=<
\let\gt=>
\def\processVert{\ifmmode|\else\textbar\fi}
\let\processvert\processVert

\@ifundefined{subparagraph}{
\def\subparagraph{\@startsection{paragraph}{5}{2\parindent}{0ex plus 0.1ex minus 0.1ex}%
{0ex}{\normalfont\small\itshape}}%
}{}

% These are now gobbled, so won't appear in the PDF.
\newcommand\role[1]{\unskip}
\newcommand\aucollab[1]{\unskip}
  
\@ifundefined{tsGraphicsScaleX}{\gdef\tsGraphicsScaleX{1}}{}
\@ifundefined{tsGraphicsScaleY}{\gdef\tsGraphicsScaleY{.9}}{}
% To automatically resize figures to fit inside the text area
\def\checkGraphicsWidth{\ifdim\Gin@nat@width>\linewidth
	\tsGraphicsScaleX\linewidth\else\Gin@nat@width\fi}

\def\checkGraphicsHeight{\ifdim\Gin@nat@height>.9\textheight
	\tsGraphicsScaleY\textheight\else\Gin@nat@height\fi}

\def\fixFloatSize#1{}%\@ifundefined{processdelayedfloats}{\setbox0=\hbox{\includegraphics{#1}}\ifnum\wd0<\columnwidth\relax\renewenvironment{figure*}{\begin{figure}}{\end{figure}}\fi}{}}
\let\ts@includegraphics\includegraphics

\def\inlinegraphic[#1]#2{{\edef\@tempa{#1}\edef\baseline@shift{\ifx\@tempa\@empty0\else#1\fi}\edef\tempZ{\the\numexpr(\numexpr(\baseline@shift*\f@size/100))}\protect\raisebox{\tempZ pt}{\ts@includegraphics{#2}}}}

%\renewcommand{\includegraphics}[1]{\ts@includegraphics[width=\checkGraphicsWidth]{#1}}
\AtBeginDocument{\def\includegraphics{\@ifnextchar[{\ts@includegraphics}{\ts@includegraphics[width=\checkGraphicsWidth,height=\checkGraphicsHeight,keepaspectratio]}}}

\DeclareMathAlphabet{\mathpzc}{OT1}{pzc}{m}{it}

\def\URL#1#2{\@ifundefined{href}{#2}{\href{#1}{#2}}}

%%For url break
\def\UrlOrds{\do\*\do\-\do\~\do\'\do\"\do\-}%
\g@addto@macro{\UrlBreaks}{\UrlOrds}



\edef\fntEncoding{\f@encoding}
\def\EUoneEnc{EU1}
\makeatother
\def\floatpagefraction{0.8} 
\def\dblfloatpagefraction{0.8}
\def\style#1#2{#2}
\def\xxxguillemotleft{\fontencoding{T1}\selectfont\guillemotleft}
\def\xxxguillemotright{\fontencoding{T1}\selectfont\guillemotright}

\newif\ifmultipleabstract\multipleabstractfalse%
\newenvironment{typesetAbstractGroup}{}{}%

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\AtBeginDocument{\let\citet\cite\let\citep\cite}
\usepackage[numbers,sort&compress,numbers,sort&compress]{natbib}


\AtBeginDocument{\let\citet\cite\let\citep\cite}
\newif\ifnokeywords\nokeywordsfalse
\def\abbrvJournalTitle{Future J. Pharm. Health. Sci.}
\def\jrnlurlname{fjphs}
\usepackage{balance}

\makeatletter\input{size10-pointfive.clo}\makeatother%
\definecolor{kwdboxcolor}{RGB}{242,242,242}
\usepackage[hidelinks,colorlinks=true,allcolors=blue]{hyperref}
\linespread{1}                 
\def\floatpagefraction{0.8}

\usepackage[paperheight=11.69in,paperwidth=8.26in,top=1in,bottom=1in,left=1in,right=.75in,headsep=24pt]{geometry}

\usepackage{multirow-custom}

\makeatletter

\def\hlinewd#1{%
  \noalign{\ifnum0=`}\fi\hrule \@height #1%
  \futurelet\reserved@a\@xhline}
\def\tbltoprule{\hlinewd{1pt}\\[-14pt]}
\def\tblbottomrule{\noalign{\vspace*{6pt}}\hline\noalign{\vspace*{2pt}}}
\def\tblmidrule{\hline\noalign{\vspace*{2pt}}}



\let\@articleType\@empty
\let\@journalDoi\@empty
\let\@journalVolume\@empty
\let\@journalIssue\@empty
\let\@crossMarkLink\@empty
\let\@receivedDate\@empty
\let\@acceptedDate\@empty
\let\@revisedDate\@empty
\let\@publishedDate\@empty
\let\@copyrightYear\@empty
\let\@firstPage\@empty
\let\@eISSN\@empty
\let\@pISSN\@empty
\def\eISSN#1{\gdef\@eISSN{#1}}
\def\pISSN#1{\gdef\@pISSN{#1}}
\def\articleType#1{\gdef\@articleType{#1}}
\def\journalDoi#1{\gdef\@journalDoi{#1}}
\def\crossMarkLink#1{\gdef\@crossMarkLink{#1}}
\def\receivedDate#1{\gdef\@receivedDate{#1}}
\def\acceptedDate#1{\gdef\@acceptedDate{#1}}
\def\revisedDate#1{\gdef\@revisedDate{#1}}
\def\publishedDate#1{\gdef\@publishedDate{#1}}
\def\copyrightYear#1{\gdef\@copyrightYear{#1}}
\def\journalVolume#1{\gdef\@journalVolume{#1}}
\def\journalIssue#1{\gdef\@journalIssue{#1}}
\def\firstPage#1{\gdef\@firstPage{#1}}


\def\author#1{%
  \gdef\@author{%
    \hskip-\dimexpr(\tabcolsep)\hskip5pt%
    \parbox{\dimexpr\textwidth-1pt}%
    {\fontsize{11}{13}\selectfont\raggedright #1}%
  }%
}


\usepackage{pharmascope-abs}
\usepackage{caption}
\usepackage{lastpage}
\usepackage{fancyhdr}
\usepackage[noindentafter,explicit]{titlesec}
\usepackage{fontspec}

\setmainfont[%
BoldFont=cambriab.ttf,%
ItalicFont=CAMBRIAI.TTF,%
BoldItalicFont=CAMBRIAZ.TTF]{Cambria.ttf}

\usepackage{qrcode}

%\def\abbrvJournalTitle{Future J. Pharm. Health. Sci.}
\def\jshort{fjphs}

\def\title#1{%
  \gdef\@title{%
%      \vspace*{-20pt}%
       \vspace*{-48pt}
       \ifx\@articleType\@empty\else{\fontsize{10}{12}\scshape\selectfont\hspace{8pt}\@articleType\hfill\mbox{}\par\vspace{2pt}}\fi%
       \minipage{\linewidth}
         \hrulefill\\[-0.7pt]%
         \parbox{.1\linewidth}{\includegraphics[width=4.2cm]{pharmasprings_logo.png}}\hfill
%         \fcolorbox{kwdboxcolor}{kwdboxcolor}{\parbox{.792\linewidth}{%
         \fcolorbox{kwdboxcolor}{kwdboxcolor}{\parbox{.729\linewidth}{%
          \begin{center}
          \vskip-4pt 
          \scshape\fontsize{18}{20}\selectfont\journalTitle\hfill\end{center}%
          \vspace*{-16pt}{\vspace*{-3.1pt}\fontsize{8}{9}\selectfont Published by Pharma Springs Publication\hfill  Journal Home Page: \href{https://pharmasprings.com/index.php/\jrnlurlname}{\color{blue}\underline{\smash{https://pharmasprings.com/\jrnlurlname}}}}\hspace*{4pt}\mbox{}}}%
       \par\rule{\linewidth}{1.3pt}%
       \endminipage%
    \par\vspace*{9.2pt}\parbox{.98\linewidth}{\linespread{.9}\raggedright\fontsize{14}{17}\selectfont #1}%
    \vspace*{-8pt}%
  }
}


\setlength{\parindent}{0pt}
\setlength{\parskip}{0.4pc plus 1pt minus 1pt}


\fancypagestyle{headings}{%
  \renewcommand{\headrulewidth}{0pt}%
  \renewcommand{\footrulewidth}{0.3pt}
  \fancyhf{}%
  \fancyhead[R]{%
    \fontsize{9.12}{11}\selectfont\RunningAuthor,\ \abbrvJournalTitle\ \@copyrightYear;\space\ifx\@journalVolume\@empty X\else\@journalVolume\fi%
    \ifx\@journalIssue\@empty\else(\@journalIssue)\fi%
    :\ \ifx\@firstPage\@empty 1\else\@firstPage\fi-\pageref*{LastPage}%
  }%
  \fancyfoot[RE]{\fontsize{9.12}{11}\selectfont\textcopyright\ Pharma Springs Publication\ $|$\space \journalTitle}%
  \fancyfoot[LO]{\fontsize{9.12}{11}\selectfont\textcopyright\ Pharma Springs Publication\ $|$\space \journalTitle}%
  \fancyfoot[RO,LE]{\fontsize{9.12}{11}\selectfont\thepage}
}\pagestyle{headings}


                                                         
\fancypagestyle{plain}{%
  \renewcommand{\headrulewidth}{0pt}%
  \renewcommand{\footrulewidth}{0.3pt}%
  \fancyhf{}%
  \fancyhead[R]{%
    \fontsize{9.12}{11}\selectfont\RunningAuthor,\ \abbrvJournalTitle\ \@copyrightYear;\space\ifx\@journalVolume\@empty X\else\@journalVolume\fi%
    \ifx\@journalIssue\@empty\else(\@journalIssue)\fi%
    :\ \ifx\@firstPage\@empty 1\else\@firstPage\fi-\pageref*{LastPage}%
  }%
  \fancyfoot[RE]{\fontsize{9.12}{11}\selectfont\textcopyright\ Pharma Springs Publication\ $|$\space \journalTitle}%
  \fancyfoot[LO]{\fontsize{9.12}{11}\selectfont\textcopyright\ Pharma Springs Publication\ $|$\space \journalTitle}%
  \fancyfoot[RO,LE]{\fontsize{9.12}{11}\selectfont\thepage}
  \ifx\@firstPage\@empty\else\setcounter{page}{\@firstPage}\fi
}


\def\NormalBaseline{\def\baselinestretch{1.1}}
\usepackage{textcase}
\setcounter{secnumdepth}{0}

\titleformat{\section}[block]{\bfseries\boldmath\NormalBaseline\filright\fontsize{10.5}{13}\selectfont}
{\thesection}
{6pt}
{\MakeTextUppercase{#1}}
[]
\titleformat{\subsection}[block]{\bfseries\boldmath\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubsection}
{6pt}
{#1}
[]
\titleformat{\subsubsection}[block]{\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubsubsection}
{6pt}
{#1}
[]
\titleformat{\paragraph}[block]{\NormalBaseline\filright\fontsize{10.5}{10}\selectfont}
{\theparagraph}
{6pt}
{#1}
[]
\titleformat{\subparagraph}[block]{\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubparagraph}
{6pt}
{#1}
[]

\titlespacing{\section}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subsection}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subsubsection}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\paragraph}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subparagraph}{0pt}{.5\baselineskip}{.5\baselineskip}

\captionsetup[figure]{skip=1.4pt,font=bf,labelsep=colon,justification=raggedright,singlelinecheck=false}
\captionsetup[table]{skip=1.4pt,font=bf,labelsep=colon,justification=raggedright,singlelinecheck=false}

\def\bibyear#1{#1}
\def\bibjtitle#1{#1}
\def\bibauand{}
\setlength\bibsep{3pt}
\setlength\bibhang{8pt}
\lefthyphenmin = 3
\makeatother
\date{}
\makeatletter
\usepackage{mdframed}
%\usepackage{balance}
\definecolor{unitednationsblue}{RGB}{222,234,246}
\definecolor{cpboxcolor}{RGB}{211,211,211}
\definecolor{citeboxcolor}{RGB}{242,242,242}
\def\printBio{%
  %\noindent\begin{mdframed}[innerbottommargin=7pt,backgroundcolor=unitednationsblue,hidealllines=true,frametitle={ABOUT AUTHORS},innerleftmargin=7pt,leftmargin=2pt,skipabove=1pt,innertopmargin=2pt,innerbottommargin=1pt,skipbelow=1pt]%
  \noindent\begin{mdframed}[innerbottommargin=7pt,backgroundcolor=unitednationsblue,hidealllines=true,frametitle={ABOUT AUTHORS},innerrightmargin=20pt,rightmargin=4,innerleftmargin=7pt,leftmargin=2pt,skipabove=1pt,innertopmargin=2pt,innerbottommargin=1pt,skipbelow=1pt]%
   \@tempcnta=0
    \loop
      \advance \@tempcnta by 1
      \def\aubioCnt{\the\@tempcnta}
      \vskip1pt
%      \noindent\hspace{-2pt}%\fbox{
%     \vspace*{1pt}
      \raisebox{62pt}%{\begin{minipage}[t]{.35\linewidth}%
      \expandafter\ifx\csname authorImage\aubioCnt\endcsname\relax%
       \else%
         \includegraphics[height=60pt,width=60pt]{\csname authorImage\aubioCnt\endcsname}%height=145pt
       \fi%
      %\end{minipage}
      %}%}
      \hskip 4pt%\fbox{
      \raisebox{52pt}{\begin{minipage}[t]{.74\linewidth}%
        {\fontsize{8}{10}\selectfont{\csname authorName\aubioCnt\endcsname\\[5pt]}}
        \fontsize{8}{10}\selectfont\csname authorDetails\aubioCnt\endcsname\\
      \end{minipage}}
%      }
%     \vskip10pt%
    \ifnum\@tempcnta < \theaubio
    \repeat
%     \vspace*{-16.5pt}%
  \end{mdframed}
}


\makeatother


\def\citeauthor{~Krishnaveni~Manubolu,
              ~Sai~Yamini~J,
              ~Charishma~R,
              ~Neelesh~Kumar~P,
              ~Kishori~O,
              ~Tanusha~K,
              ~Kudipudi~Harinadha~Baba}
\def\citetitle{\textbf{Formulation and Evaluation of Phytosome and Plants Exhibiting Anti-Inflammatory Activity}}
\def\copyrightstmt{This is an open access article distributed under the terms of the Creative Commons Attribution-NonCommercial- ShareAlike 4.0 License, which allows others to remix, tweak, and build upon the work non-commercially, as long as the author is credited and the new creations are licensed under the identical terms.}
\publishedDate{}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% Feature enabled:
%citation-style: unsrt
%citation-type: numbered
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\usepackage{float}

\begin{document}


\def\RunningAuthor{Krishnaveni Manubolu et al.}
\firstPage{90}
\articleType{Original Article}
\receivedDate{10 Dec 2022}
\acceptedDate{04 Jan 2023}
\revisedDate{02 Jan 2023}
\journalVolume{3}
\journalIssue{1}
\journalDoi{fjphs.v3i1.351}
\copyrightYear{2023}
\eISSN{2583-116X}
\journalDoi{fjphs.v3i1.351}
\def\authorCount{7}
\def\affCount{3}

\def\journalTitle{Future Journal of Pharmaceuticals and Health Sciences}

\title{\textbf{Formulation and Evaluation of Phytosome and Plants Exhibiting Anti-Inflammatory Activity}}
\author{Krishnaveni~Manubolu\textsuperscript{*}\textsuperscript{1}\textsuperscript{\href{https://orcid.org/0000-0002-9401-1768}{\XeTeXLinkBox{\includegraphics[height=7pt]{orcid-logo-pharmascope.pdf}}}},  
            Sai~Yamini~J\textsuperscript{2},  
            Charishma~R\textsuperscript{2},  
            Neelesh~Kumar~P\textsuperscript{2},  
            Kishori~O\textsuperscript{2},  
            Tanusha~K\textsuperscript{2}, 
            Kudipudi~Harinadha~Baba\textsuperscript{3}~\\[5pt]{\textsuperscript{1}Department of Pharmaceutics, Narayana Pharmacy College, Chinthareddy Palem, SPSR Nellore - 524 002, Andhra Pradesh, India}~\\{\textsuperscript{2}Narayana Pharmacy College, Chinthareddy Palem, SPSR Nellore - 524 002, Andhra Pradesh, India}~\\{\textsuperscript{3}Department of Pharmaceutical Analysis, Narayana Pharmacy College, Chinthareddy Palem, SPSR Nellore - 524 002, Andhra Pradesh, India}}

\begin{abstract}
The aim of the present study is formulation \& evaluation of phytosome complexes from plant extracts of \textit{Ixora coccinea}, \textit{Chrysanthemum morifolium} \& \textit{Tinospora cordifolia}. Plants exhibiting anti-inflammatory activity (\textit{Ixora coccinea, Chrysanthemum morifolium, and Tinospora cordifolium}) were selected. To carry out the Cold maceration process, prepare an aqueous and alcoholic extract of \textit{Ixora coccinea, Chrysanthemum morifolium, and Tinospora cordifolia}. To perform Phytochemical screening, phytochemical evaluation, and FTIR spectral analysis. To prepare phagosomes by cholesterol complex method. Optimization of phagosomal formulations. The in vitro drug release from F1 to F9 was preserved in table below. It is executed that 8-hour drug releases time profile From the results, it was reported that F1 formulation preparation using \textit{Ixora coccinea} exhibited 19.9\% of drug release, which is less than 30\%, indicating that worst release was not observed. This is the desired characteristic to be passed by the control release formula. At the end of the 4th hour, 48.9 \% of drug release was observed 98.7\% of Drug was released at the end of the 8th hour. 
\end{abstract}\def\keywordstitle{Keywords}

\begin{keywords}Phytosome,\newline Anti-Inflammatory,\newline Ixora coccinea
\end{keywords}
\twocolumn[ \maketitle {\printKwdAbsBox}]
\makeatletter{\raggedright{\Large$^\ast$}Corresponding Author\par Name:\ Krishnaveni~Manubolu~\\ Phone:\ +91 8074866545~\\  Email:\ krishnaveni.manubolu@gmail.com
        \par}\vspace*{-13pt}\hrulefill\par\ifx\@eISSN\@empty\else{\fontsize{12}{14}\selectfont eISSN: \@eISSN\hfill pISSN: \@pISSN}\par\fi% 
        \textsc{DOI:}\ \href{https://doi.org/10.26452/\@journalDoi}{\textcolor{blue}{\underline{\smash{https://doi.org/10.26452/\@journalDoi}}}}\par%
        \vspace*{-12pt}\hrulefill\\[5pt]\hypersetup{allcolors=black}\qrcode[version=4,height=1.5cm]{https://doi.org/10.26452/\@journalDoi}\hypersetup{allcolors=blue} \hspace*{10.5pt}\parbox[t]{5cm}{\raisebox{19pt}{\fontsize{9.12}{10.12}\selectfont Production and Hosted by}\\[-13pt]{{\fontsize{12}{14}\selectfont  Pharmasprings.com \\[-1pt]{\fontsize{9.12}{10.12}\selectfont\textcopyright\ \@copyrightYear\ $|$ All rights reserved.}\\[-8pt]\hspace*{-1pt}\rule{.762\columnwidth}{1.2pt}}}}\hypersetup{allcolors=blue}%{\fontsize{12}{14}\selectfont Pharmasprings.com}\par%
        \makeatother
        
    
\section{Introduction}
During oral and topical applications, pyrosomes enhance lipid-insoluble polarity phytoconstituent absorption, exhibiting increased bioavailability and a significantly elevated therapeutic benefit. The quantity of active compounds that must be consumed typically reduces as their rate for absorption increases \unskip~\cite{1640535:26773296}. Flavonoid and terpenoid compounds discovered in plant preparations were well suited toward direct binding with phosphatidylcholine. A standardized extraction, as well as polyphenolic constituents (including such simple flavonoids), were mixed with a stoichiometric amount of phospholipid (phosphatidylcholine) to generate phytosomes \unskip~\cite{1640535:26773297}. A lipophilic phosphatidyl moiety and an amino group make up a bifunctional compound, phosphatidylcholine. A hydrophilic moiety of choline. The choline head of the phosphatidylcholine molecule selectively attaches to these substances, and the body or tail of a lipid-soluble phosphatidyl section after that wraps this choline-bound substance. Damage is caused by gut microorganisms but also stomach secretions \unskip~\cite{1640535:26773287}.

As a response, these phytoconstituents create a biochemical complex that's also biocompatible with lipids along with phospholipids, known as the phyto-phospholipid complex. Specific spectroscopic techniques show that molecules are anchored to the polar choline head of the phospholipids via chemical bonds. According to precise chemical analysis, the unit phytosome is typically a flavonoid molecule linked to at least one phosphatidylcholine molecule. As a result, a small microsphere or cell is formed. The phytosome technology creates a small enclosure that protects the plant extract or its active constituent \unskip~\cite{1640535:26773289}.
    
\section{Materials and Methods}
Fresh leaves were collected from fully grown flower plants of \textit{Ixora coccinea} from Nellore, N-Hexane from NICE chemicals, Ltd., Kerala, Ethanol from Jiangsu Huaxi, Ethyl acetate from Ranbaxy fine Chemical Ltd, New Delhi, Soya lecithin is from Bakers villa, cholesterol is from S D Fine Chemicals Ltd, Mumbai, Bromine water is from Ranbaxy fine Chemical Ltd, New Delhi, and remaining reagents like Mayers reagent, Barfoeds reagent, H2So4, HCl from NICE chemicals, Ltd., Kerala.

\textbf{Preparation of Phytosome}

Phytosomes are prepared using the cholesterol complexation method \unskip~\cite{1640535:26773294}. The different molar ratio of the pyrosomes was designed by the selected optimize working method as mentioned in Table~\ref{tw-a0003659fb43}.

\textbf{Preformulation Studies }

\textbf{Organoleptic Evaluation}

The samples of plant extracts [Table~\ref{tw-afa21eb9103d}] \textit{Ixora coccinea, Tinospora cordifolia, and Chrysanthemum morifolium}, were studied for organoleptic characteristics such as color, odor, \& solubility \unskip~\cite{1640535:26773292}.

\textbf{Melting Point}

Plant extracts \textit{Ixora coccinea}, \textit{Chrysanthemum morifolium}, and \textit{Tinospora cordifolia} has been established using such melting point equipment as well as the capillary technique \unskip~\cite{1640535:26773285}.

\textbf{FTIR Spectroscopy}

Using a Shimadzu-8400S FT-IR Spectrophotometer, the medications' ft-or range was obtained (Tokyo, Japan). Each blend of Ixora +cholesterol, Ixora +soya lecithin, Tinospora + cholesterol, and chrysanthemum +cholesterol was compacted to a semisolid paste. 

They were scanned using a Fourier transform infrared instrument inside a wavenumber range of 3500 to 1000 cm, and spectral analysis was performed. The software was employed to analyze the data \unskip~\cite{1640535:26773299}  (UV PROB version 14).

\textbf{Determination of Solubility}

The Plant extracts \textit{Chrysanthemum morifolium, Tinospora cordifolia, and Ixora coccinea} were determined in various organic solvents like ethyl acetate, ethanol, dichloro methane, DMSO, and distilled water was done \unskip~\cite{1640535:26773290}.

\textbf{UV Spectroscopy Study (Determination of \ensuremath{\lambda }m max) }

The standard stock solution of 50\ensuremath{\mu}g/ml of \textit{drug } \textit{Ixora coccinea, Chrysanthemum morifolium, and Tinospora cordifolia }was filtered between 200 -800 nm using a UV spectrophotometer in water \& ethanol, respectively.

\textbf{Evaluation of Phytosomes}

The phytosomes of plant extract were characterized by FTIR and Calibration studies \unskip~\cite{1640535:26773293,1640535:26773291,1640535:26773288} and assessed for percent yield, percent entrapment efficiency, particle size, and in vitro drug release  \unskip~\cite{1640535:26773286,1640535:26773295,1640535:26773298}.

\textbf{Optimization of a Final Batch of Phytosomes}

Based on in-vitro dissolution parameters, formulations were optimized.\ensuremath{^{}}

\textbf{Determination of \% Yield}

The simultaneous equation calculated assurance of \% yield of formulations: 

$\style{font-family:'Times New Roman'}{\style{font-size:8px}{(\%)\;Yield=\frac{Practical\;yield}{Theoretical\;yield}\times100}} $

\textbf{Determination of } \textbf{the Average Particle Size}

The results of optical microscopy were conducted to identify. The average particle size of the phytosomal formulation of \textit{Ixora coccinea, chrysanthemum morifolium \& Tinospora cordifolia} was determined.

\textbf{Drug Entrapment Efficiency}

Each of the produced phytosomes' drug entrapment effectiveness was determined as a percentage. A 100 ml volumetric flask containing 100 ml of phosphate-buffered saline (pH 6.8) with 100 mg of a product has been measured accordingly and set aside. 

A volumetric flask was stirred continuously at 35\ensuremath{^\circ} C for two hours that following day to release all medicament from formulations. 

After that, using an ultraviolet (UV) spectrophotometer, 1 ml of a solution was diluted upto 10 ml and then evaluated for drug entrapment efficiency 286, 256, 410 nm, respectively for \textit{Ixora, Chrysanthemum \& Tinospora, } respectively. Drug entrapment was calculated using the formula. 

$\%\;of\;Drug\;Entrapment\;Efficiency=\frac{Actual\;drug\;content}{theoretical\;}\times100 $

\textbf{Drug Release Kinetic Model}

Zero-order release (cumulative percent drug released vs. time) equation [Figure~\ref{f-3a9b1a7ed178}]
\let\saveeqnno\theequation
\let\savefrac\frac
\def\dispfrac{\displaystyle\savefrac}
\begin{eqnarray}
\let\frac\dispfrac
\gdef\theequation{1}
\let\theHequation\theequation
\label{dfg-b9ab31186e9d}
\begin{array}{@{}l}Q=K_0t\end{array}
\end{eqnarray}
\global\let\theequation\saveeqnno
\addtocounter{equation}{-1}\ignorespaces 
First order release (cumulative log percentage of Drug remaining vs. time) equation [Figure~\ref{f-f8aac367f7d8}]
\let\saveeqnno\theequation
\let\savefrac\frac
\def\dispfrac{\displaystyle\savefrac}
\begin{eqnarray}
\let\frac\dispfrac
\gdef\theequation{2}
\let\theHequation\theequation
\label{dfg-41939b10fa88}
\begin{array}{@{}l}Log\;Q_t\;=\;Log\;Q_0\;+\;K\;_1t/2.303\end{array}
\end{eqnarray}
\global\let\theequation\saveeqnno
\addtocounter{equation}{-1}\ignorespaces 
Higuchi's (cumulative percent drug released Vs. square root of time) equation [Figure~\ref{f-f56928b2de3f}]
\let\saveeqnno\theequation
\let\savefrac\frac
\def\dispfrac{\displaystyle\savefrac}
\begin{eqnarray}
\let\frac\dispfrac
\gdef\theequation{3}
\let\theHequation\theequation
\label{dfg-6bdff9fbd469}
\begin{array}{@{}l}Q\;=\;K_H\;t^{1/2}\end{array}
\end{eqnarray}
\global\let\theequation\saveeqnno
\addtocounter{equation}{-1}\ignorespaces 
Korsmeyer and Peppa's (Log cumulative percent drug released versus log time equation) [Figure~\ref{f-448410500c0d}]
\let\saveeqnno\theequation
\let\savefrac\frac
\def\dispfrac{\displaystyle\savefrac}
\begin{eqnarray}
\let\frac\dispfrac
\gdef\theequation{4}
\let\theHequation\theequation
\label{dfg-df72fdc4bd0a}
\begin{array}{@{}l}F\;=\;(M_t\;/\;M)\;=\;K_m\;t^{n}\;\;\end{array}
\end{eqnarray}
\global\let\theequation\saveeqnno
\addtocounter{equation}{-1}\ignorespaces 
Where,

 'Q' is the number of Drugs released at the time; 'M\ensuremath{_{t}}' is drug release at the time; 'M' is the total amount of the Drug in dosage form; 'F' is a fraction of the drug released at the time; 'K\ensuremath{_{0}}' is zero order release rate constant; 'K\ensuremath{_{H}}' is Higuchi's square root of the time release rate regular; 'K\ensuremath{_{m}}' is constant and depends on the geometry of the dosage form.
    
\section{RESULTS AND DISCUSSION}
Organoleptic Evaluation: \ensuremath{\lambda }\ensuremath{_{max}}: Absorption maxima(\ensuremath{\lambda }\ensuremath{_{max)\ }}in plant extraction of Ixora coccinea leaves [Figure~\ref{f-9964525a25fd}].

Calibration: Standard Calibration curve of Ixora coccinea in distilled water at 286nm [Figure~\ref{f-cc5b8f1de862}  and Table~\ref{tw-4481750cddf6}].

Determination Solubility Studies: Solubility of Ixora coccinea at 286nm [Table~\ref{tw-92546b5f33bf}].


\begin{table*}[!htbp]
\caption{\boldmath {Preparation of Phytosomes} }
\label{tw-a0003659fb43}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.3343\linewidth-2\tabcolsep}p{\dimexpr.33019999999999996\linewidth-2\tabcolsep}p{\dimexpr.33550000000000004\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}{\multirow{2}{*}{\cAlignHack Plant extract phytosomes}} & \multicolumn{2}{p{\dimexpr(.6657\linewidth-2\tabcolsep)}}{\cAlignHack Molar ratios}\\
\rowcolor{kwdboxcolor} & \cAlignHack Extract & \cAlignHack Cholesterol\\
\tblmidrule 
\cAlignHack F1 &
  \cAlignHack 1 &
  \cAlignHack 1\\
\cAlignHack F2 &
  \cAlignHack 1 &
  \cAlignHack 2\\
\cAlignHack F3 &
  \cAlignHack 2 &
  \cAlignHack 1\\
\cAlignHack F4 &
  \cAlignHack 1 &
  \cAlignHack 1\\
\cAlignHack F5 &
  \cAlignHack 1 &
  \cAlignHack 2\\
\cAlignHack F6 &
  \cAlignHack 2 &
  \cAlignHack 1\\
\cAlignHack F7 &
  \cAlignHack 1 &
  \cAlignHack 1\\
\cAlignHack F8 &
  \cAlignHack 1 &
  \cAlignHack 2\\
\cAlignHack F9 &
  \cAlignHack 2 &
  \cAlignHack 1\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Organoleptic Evaluation of \textit{Ixora coccinea}, \textit{Chrysanthemum morifolium} and \textit{Tinospora cordifolia}} }
\label{tw-afa21eb9103d}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.0994\linewidth-2\tabcolsep}p{\dimexpr.30970000000000002\linewidth-2\tabcolsep}p{\dimexpr.12630000000000004\linewidth-2\tabcolsep}p{\dimexpr.2646\linewidth-2\tabcolsep}p{\dimexpr.20\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}S.no  & Name of the Plant  & Colour  & Odour  & Solubility \\
\tblmidrule 
1 &
  \textit{Ixora coccinea} &
  Brown &
  Fragrant &
  Freely soluble\\
2  &
  \textit{Tinospora cordifolia} &
  Green  &
  Characteristic odor  &
  Freely soluble \\
3  &
  \textit{Chrysanthemum morifolium } &
  Yellow   &
  Fragrant  &
  Freely soluble \\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Calibration Values of \textit{Ixora coccinea} in Distilled Water at 286 nm} }
\label{tw-4481750cddf6}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.4663\linewidth-2\tabcolsep}p{\dimexpr.5337\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}\cAlignHack Concentration (\ensuremath{\mu}g/ml) & \cAlignHack Absorbance \\
\tblmidrule 
\cAlignHack 0 &
  \cAlignHack 0\\
\cAlignHack 20 &
  \cAlignHack 0.258\\
\cAlignHack 40 &
  \cAlignHack 0.502\\
\cAlignHack 60 &
  \cAlignHack 0.614\\
\cAlignHack 80 &
  \cAlignHack 0.786\\
\cAlignHack 100 &
  \cAlignHack 0.942\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Solubility of \textit{Ixora coccinea}} }
\label{tw-92546b5f33bf}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.21539999999999996\linewidth-2\tabcolsep}p{\dimexpr.348\linewidth-2\tabcolsep}p{\dimexpr.4366\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}S.no & Solvents & Solubility\\
\tblmidrule 
1 &
  Water &
  Partially soluble\\
2 &
  Ethanol &
  Partially soluble\\
3 &
  Ethyl acetate &
  Immiscible\\
4 &
  DMSO &
  Completely soluble\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Fourier {\textendash} Transform Infrared Spectroscopy Absorption of \textit{Ixora coccinea}} }
\label{tw-e4ee283e7cfe}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.1102\linewidth-2\tabcolsep}p{\dimexpr.28979999999999997\linewidth-2\tabcolsep}p{\dimexpr.20\linewidth-2\tabcolsep}p{\dimexpr.20\linewidth-2\tabcolsep}p{\dimexpr.20\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}S.no & Functional groups of \textit{Ixora coccinea} & Infrared absorption of plant extract & Infrared absorption of drug \& soya lecithin & Infrared absorption of Drugs \& cholesterol\\
\tblmidrule 
01 &
  OH {\textendash} Stretching; O-H stretching; OH {\textendash} Stretching &
  3338 cm\ensuremath{^{-1}} &
  3344 cm\ensuremath{^{-1}} &
  3338 cm\ensuremath{^{-1}}\\
02 &
  C\ensuremath{\Xi }C stretching; C\ensuremath{\Xi }C stretching; C\ensuremath{\Xi }C stretching &
  2114 cm\ensuremath{^{-1}} &
  2114 cm\ensuremath{^{-1}} &
  2114 cm\ensuremath{^{-1}}\\
03 &
  C=C stretching; C=C stretching; C=C stretching &
  1636 cm\ensuremath{^{-1}} &
  1636 cm\ensuremath{^{-1}} &
  1636 cm\ensuremath{^{-1}}\\
04 &
  S=O stretching; S=O stretching; S=O stretching &
  1407 cm\ensuremath{^{-1}} &
  1408 cm\ensuremath{^{-1}} &
  1407 cm\ensuremath{^{-1}}\\
05 &
  C-O stretching; C-O stretching; C-O stretching &
  1095 cm\ensuremath{^{-1}} &
  1085 cm\ensuremath{^{-1}} &
  1095 cm\ensuremath{^{-1}}\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Percentage Yield of Phytosomal Formulations} }
\label{tw-bcf2ea031228}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.4955\linewidth-2\tabcolsep}p{\dimexpr.5045\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}\cAlignHack Formulation & \cAlignHack Percentage yield(\%) \\
\tblmidrule 
\cAlignHack F1 &
  \cAlignHack 80\%\\
\cAlignHack F2 &
  \cAlignHack 70\%\\
\cAlignHack F3 &
  \cAlignHack 40\%\\
\cAlignHack F4 &
  \cAlignHack 50\%\\
\cAlignHack F5 &
  \cAlignHack 46\%\\
\cAlignHack F6 &
  \cAlignHack 53\%\\
\cAlignHack F7 &
  \cAlignHack 40\%\\
\cAlignHack F8 &
  \cAlignHack 33\%\\
\cAlignHack F9 &
  \cAlignHack 40\%\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Drug Entrapment Efficiency} }
\label{tw-287adf851b2d}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.4978\linewidth-2\tabcolsep}p{\dimexpr.5022\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}\cAlignHack Formulation code & \cAlignHack \% Entrapment efficiency\\
\tblmidrule 
\cAlignHack F1 &
  \cAlignHack 90.90\\
\cAlignHack F2 &
  \cAlignHack 85.91\\
\cAlignHack F3 &
  \cAlignHack 83.31\\
\cAlignHack F4 &
  \cAlignHack 81.17\\
\cAlignHack F5 &
  \cAlignHack 83.64\\
\cAlignHack F6 &
  \cAlignHack 90.01\\
\cAlignHack F7 &
  \cAlignHack 92.91\\
\cAlignHack F8 &
  \cAlignHack 72.34\\
\cAlignHack F9 &
  \cAlignHack 76.79\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Invitro Drug Release Studies} }
\label{tw-1a23d309cc36}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.0982\linewidth-2\tabcolsep}p{\dimexpr.1018\linewidth-2\tabcolsep}p{\dimexpr.10\linewidth-2\tabcolsep}p{\dimexpr.10\linewidth-2\tabcolsep}p{\dimexpr.10\linewidth-2\tabcolsep}p{\dimexpr.10\linewidth-2\tabcolsep}p{\dimexpr.10\linewidth-2\tabcolsep}p{\dimexpr.10\linewidth-2\tabcolsep}p{\dimexpr.10\linewidth-2\tabcolsep}p{\dimexpr.10\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}Time (min) & F1 & F2 & F3 & F4 & F5 & F6 & F7 & F8 & F9\\
\tblmidrule 
1 &
  11.5 &
  20.6 &
  12.6 &
  36.9 &
  38.1 &
  12.3 &
  13.5 &
  10.3 &
  13.3\\
2 &
  19.9 &
  31.5 &
  19.7 &
  39.4 &
  42.5 &
  24.6 &
  27.7 &
  17.1 &
  18.9\\
3 &
  34.7 &
  39.6 &
  33.8 &
  45.3 &
  45.6 &
  38.2 &
  39.1 &
  28.4 &
  31.4\\
4 &
  48.9 &
  42.3 &
  46.8 &
  49.7 &
  51.1 &
  40.2 &
  42.9 &
  30.5 &
  40.6\\
5 &
  60.4 &
  58.4 &
  57.2 &
  61.2 &
  55.6 &
  53.9 &
  54.5 &
  52.7 &
  53.8\\
6 &
  72.5 &
  71.5 &
  70.8 &
  66.4 &
  61.3 &
  75.9 &
  68.3 &
  64.2 &
  67.2\\
7 &
  83.2 &
  80.5 &
  80.6 &
  69.1 &
  65.4 &
  89.5 &
  74.6 &
  75.2 &
  76.5\\
8 &
  98.7 &
  92.3 &
  96.5 &
  70.1 &
  78.9 &
  92.5 &
  80.2 &
  82.5 &
  81.3\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Data of Drug Release {\textendash} Kinetics of Optimized Formulation} }
\label{tw-c4e918a2413a}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.091\linewidth-2\tabcolsep}p{\dimexpr.109\linewidth-2\tabcolsep}p{\dimexpr.10\linewidth-2\tabcolsep}p{\dimexpr.10\linewidth-2\tabcolsep}p{\dimexpr.1126\linewidth-2\tabcolsep}p{\dimexpr.0874\linewidth-2\tabcolsep}p{\dimexpr.10\linewidth-2\tabcolsep}p{\dimexpr.0829\linewidth-2\tabcolsep}p{\dimexpr.14590000000000003\linewidth-2\tabcolsep}p{\dimexpr.07119999999999999\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}Time (Hr) & Cumulative \% drug released & \% Drug remaining & Square root time & Log Cumu \% drug remaining & Log time & Log Cumu \% drug released & \% Drug releas-   ed & Cube Root of \% drug Remaining(Wt) & Wo-Wt\\
\tblmidrule 
0 &
  0 &
  100 &
  0.000 &
  2.000 &
  0.000 &
  0.000 &
  100 &
  4.642 &
  0.000\\
1 &
  11.5 &
  88.5 &
  1.000 &
  1.947 &
  0.000 &
  1.061 &
  11.5 &
  4.456 &
  0.186\\
2 &
  19.9 &
  80.1 &
  1.414 &
  1.904 &
  0.301 &
  1.299 &
  8.4 &
  4.311 &
  0.331\\
3 &
  34.7 &
  65.3 &
  1.732 &
  1.815 &
  0.477 &
  1.540 &
  14.8 &
  4.027 &
  0.615\\
4 &
  48.9 &
  51.1 &
  2.000 &
  1.708 &
  0.602 &
  1.689 &
  14.2 &
  3.711 &
  0.931\\
5 &
  60.4 &
  39.6 &
  2.236 &
  1.598 &
  0.699 &
  1.781 &
  11.5 &
  3.409 &
  1.233\\
6 &
  72.5 &
  27.5 &
  2.449 &
  1.439 &
  0.778 &
  1.860 &
  12.1 &
  3.018 &
  1.624\\
7 &
  83.2 &
  16.8 &
  2.646 &
  1.225 &
  0.845 &
  1.920 &
  10.7 &
  2.561 &
  2.081\\
8 &
  98.7 &
  1.3 &
  2.828 &
  0.114 &
  0.903 &
  1.994 &
  15.5 &
  1.091 &
  3.551\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\bgroup
\fixFloatSize{images/78902c9c-14d5-4df0-a9db-4f45885b4ddd-upicture1.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/78902c9c-14d5-4df0-a9db-4f45885b4ddd-upicture1.png}{\includegraphics{images/78902c9c-14d5-4df0-a9db-4f45885b4ddd-upicture1.png}}{}
\makeatother 
\caption{\boldmath {The \ensuremath{\lambda }\ensuremath{_{max\ }}for the Plant Extract of \textit{Ixora coccinea} was Found to be at 286 nm}}
\label{f-9964525a25fd}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/a47323b1-5ed6-49b9-a9f5-27536895a6c5-upicture2.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/a47323b1-5ed6-49b9-a9f5-27536895a6c5-upicture2.png}{\includegraphics{images/a47323b1-5ed6-49b9-a9f5-27536895a6c5-upicture2.png}}{}
\makeatother 
\caption{\boldmath {Standard Calibration Curve of \textit{Ixora coccinea} in Distilled Water at 286 nm}}
\label{f-cc5b8f1de862}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/4b38a5f0-41e2-4d67-a6e8-63a4edac2b19-upicture3.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/4b38a5f0-41e2-4d67-a6e8-63a4edac2b19-upicture3.png}{\includegraphics{images/4b38a5f0-41e2-4d67-a6e8-63a4edac2b19-upicture3.png}}{}
\makeatother 
\caption{\boldmath {Fourier {\textendash}Transform Infrared Spectroscopy Absorption of \textit{Ixora coccinea}}}
\label{f-8641beb1a8b6}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/a22f542f-74d8-4188-b1eb-6684babae044-upicture4.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/a22f542f-74d8-4188-b1eb-6684babae044-upicture4.png}{\includegraphics{images/a22f542f-74d8-4188-b1eb-6684babae044-upicture4.png}}{}
\makeatother 
\caption{\boldmath {Zero-Order Release (Cumulative Percent Drug Released vs. Time)}}
\label{f-3a9b1a7ed178}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/9d3067dd-c6a8-4a07-9687-5c2cf6d8132e-upicture5.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/9d3067dd-c6a8-4a07-9687-5c2cf6d8132e-upicture5.png}{\includegraphics{images/9d3067dd-c6a8-4a07-9687-5c2cf6d8132e-upicture5.png}}{}
\makeatother 
\caption{\boldmath {First Order Release (Cumulative Log Percentage of Drug Remaining vs. Time)}}
\label{f-f8aac367f7d8}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/e92f3f96-e7a1-4de4-9f42-4fa3fe2cfd9c-upicture6.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/e92f3f96-e7a1-4de4-9f42-4fa3fe2cfd9c-upicture6.png}{\includegraphics{images/e92f3f96-e7a1-4de4-9f42-4fa3fe2cfd9c-upicture6.png}}{}
\makeatother 
\caption{\boldmath {Korsmeyer and Peppa's (Log Cumulative Percent Drug Released vs. Log Time)}}
\label{f-448410500c0d}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/f7fb1b12-baa7-454b-bf32-7c587dce0c21-upicture7.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/f7fb1b12-baa7-454b-bf32-7c587dce0c21-upicture7.png}{\includegraphics{images/f7fb1b12-baa7-454b-bf32-7c587dce0c21-upicture7.png}}{}
\makeatother 
\caption{\boldmath {Higuchi's (Cumulative Percent Drug Released vs. Square Root of Time)}}
\label{f-f56928b2de3f}
\end{figure}
\egroup
\textbf{FTIR Interpretations}

From the FTIR spectral analysis of \textit{Tinospora cordifolia} extract and \textit{Tinospora cordifolia} excipients and \textit{chrysanthemum morifolium} and \textit{chrysanthemum morifolium} extract + excipients. Presented shifts of valence vibration bands of flavonoids carboxyl group according to the number of -OH groups and strong bonds relating to a characteristic of aromatic ring vibration appeared in the range of 1448.95 - 1449.50 Cm\ensuremath{^{-1}}. 

And also indicated the presence of C-OH stretching Nitrations. After observation of results in comparison with plant extract \& Plant extract + excipients didn't exhibit new peaks appearance and existing peaks disappearance. It showed good compatibility between plant extract and excipients [Figure~\ref{f-8641beb1a8b6}  \& Table~\ref{tw-e4ee283e7cfe}].

\textbf{Determination of Percentage Yield }

The following equation was used to calculate the assurance of the percentage yield of formulations [Table~\ref{tw-bcf2ea031228}].

$\style{font-family:'Times New Roman'}{\style{font-size:8px}{\%Yield=\frac{Practical\;yield}{Theoretical\;yield}\times100}} $

\textbf{Determination of Particle Size}

Optical microscopy was used to determine particle size, and the particle size of ixora coccinea formulations was 63\ensuremath{\mu }m. Optical microscopy was used to determine particle size, and the size of chrysanthemum morifolium fomulations was found to be 56\ensuremath{\mu }m. 

The particle size of \textit{Tinospora cardifolia} formulations was determined using optical microscopy and was 76\ensuremath{\mu }m.

\textbf{Drug Entrapment Efficiency}

Virtues for such efficiency of drug entrapment within produced phytosomes with \textit{Ixora coccinea, Chrysanthemum morifolium, and Tinospora cardifolia}, F1 to F9, are shown in the above table and thus are found appropriate for each formulation [Table~\ref{tw-287adf851b2d}]. 

Therefore, formulations developed by this provide superior value cholestrol complexation {\textendash} F1 to F3.

\textbf{Invitro Drug Release Studies}

The in vitro drug release from F1 to F9 was preserved in Table~\ref{tw-1a23d309cc36}. It is executed that 8-hour drug releases time profile From the results, it was reported that F1 formulation preparation using Ixora coccinea exhibited 19.9\% of drug release, less than 30\% indicating that worst release was not observed. 

This is the desired characteristic to be passed by the control release formula. At the end of the 4th hour, 48.9 \% of drug release was observed 98.7\% of Drug was released at the end of the 8th hour.

\textbf{Release Kinetics}

The Drug's release from phytosomes indicates the mechanism of hixon drug release (R2 = 0.9976), which involves dissolution factors from the amphiphilic nature of cholesterol. The Information from dissolution experiments have been modelled using a variety of Kinetics equation the results of kinetic models was present in Table~\ref{tw-c4e918a2413a}. The data were plotted according to the mathematical models results exhibited that optimized phytosomes followed zero order kinetics with regression value (0.9993).

\textbf{Optimization of Phytosomes}

Phytosomes (F1) formulation showed better percentage yield, EE, particle size, and drug release than other phytosomes. From the results, F1 was selected as optimized phytosomes.
    
\section{CONCLUSION}
Numerous chronic diseases are thought by experts to be influenced by inflammation. An illustration of this would be the metabolic syndrome, which encompasses diabetes, type 2 cardiovascular disease, and being overweight. People with these illnesses typically have higher concentrations or markers of inflammation within their organs. A body's natural defense response involves inflammation, which further assists recovery. A foreign substance, such as a thorn, an irritation, or even an infection, may be the assailant. Pathogens usually cause diseases, including bacteria, viruses, and other creatures. Your body sometimes perceives its very own tissues or cells as harmful. Autoimmunity problems such as type 1 diabetes may develop due to such a reaction. According to specialists, inflammation might play a part in various chronic illnesses. This metabolic syndrome mainly encompasses type two diabetes, cardiovascular disease, and overweight is an instance of all this. Inflammatory mediators were typically found in higher concentrations inside the bodies of individuals with all these disorders.
    
\section{Acknowledgement}
I want to thank my esteemed Principal (K. Harinadha Baba, sir) \& My guide, Dr. S. Sujatha Madam, Professor \& Head of Department of Pharmaceutics, Narayana Pharmacy College, Chinthareddypalem, SPSR Nellore-524 002, Andhra Pradesh, India \& For her encouragement and kind suggestions to carry out my research work successfully.

\textbf{Conflict of Interest}

The authors declare no conflict of interest, financial or otherwise.

\textbf{Funding Support}

The authors declare that they have no funding support for this study.
    

\bibliographystyle{unsrt}

\bibliography{\jobname}\vspace*{20pt} 
\begin{mdframed}[backgroundcolor=cpboxcolor,hidealllines=true,innerbottommargin=7pt,innerrightmargin=1pt,rightmargin=1pt,innerleftmargin=1pt,leftmargin=2pt,skipabove=1pt,innertopmargin=2pt,innerbottommargin=1pt,skipbelow=1pt]%
{\fontsize{8.5}{10}\selectfont\textbf{Copyright:}~\copyrightstmt}	
\end{mdframed}
        
\begin{mdframed}[backgroundcolor=citeboxcolor,hidealllines=true,innerbottommargin=7pt,innerrightmargin=4pt,rightmargin=4pt,innerleftmargin=2pt,leftmargin=2pt,skipabove=1pt,innertopmargin=2pt,innerbottommargin=4pt,skipbelow=2pt]%
{\fontsize{8.5}{10}\selectfont\textbf{Cite~this~article:}~\citeauthor.~\citetitle.\space\makeatletter\abbrvJournalTitle\ \@copyrightYear;\ \ifx\@journalVolume\@empty X\else\@journalVolume\fi\ifx\@journalIssue\@empty\else(\@journalIssue)\fi:\ \ifx\@firstPage\@empty 1\else\@firstPage\fi-\pageref*{LastPage}.\makeatother}	
\end{mdframed}
\vskip-1.1pc

\vskip.5pc
{\includegraphics[width=5cm]{pharmasprings_logo.png}}
~\\
{\fontsize{9.12}{10.12}\selectfont\textcopyright\ \makeatletter\@copyrightYear\makeatother\ Pharma Springs Publication.}

\end{document}
